Author's response to reviews

Title: Exploratory trials, confirmatory observations: a new reasoning model in the era of patient-centered medicine

Authors:

Jose A Sacristan (sacristan_jose@lilly.com)

Version: 3 Date: 10 April 2011

Author's response to reviews:

Dear Dr Aldcroft,

Please find enclosed the new version of the paper entitled ‘Exploratory trials, confirmatory observations: A new reasoning model in the era of patient-centered medicine”. It includes all the suggestions made by the reviewer.

Yours faithfully,

José A Sacristán
Head of Clinical Research Department, Lilly Spain